Nuclear Dependence of Drell-Yan from E866/NuSea ffl Dominated by Shadowing with little room for Initial State

Energy Loss.

ffl Cutting out Shadowing Region (x2 > 0.035) leaves no signif-

icant nuclear effect.

ffl Preliminary Limits on quark energy loss (from our data) in

the context of three models – Gavin & Milana, PRL 68, 1834 (1992): ΔE/E < 0.13%/fm � L or < 1.6 GeV for W – Brodsky & Hoyer, PL B 298, 1645 (1993): ΔE < 0.41 GeV/fm � L or < 1.9 GeV for W – Baier et al., NP B 484, 265 (1997) & HEP-PH/9804212:

ΔE < 0.043 GeV/fm^2 � L^2 or < 1.4 GeV for W